Author's response to reviews

Title: Evaluation of non-inferiority of intradermal versus adjuvanted seasonal influenza vaccine using two serological techniques: a randomised comparative study

Authors:

Pierre Van Damme (pierre.vandamme@ua.ac.be)
Robert Arnou (alti2@wanadoo.fr)
Froukje Kafejea (froukje.oosterhuis@ua.ac.be)
Anne Fiquet (afiquet@spmsd.com)
Patrick Richard (prichard@spmsd.com)
Stephane Thomas (sthomas@spmsd.com)
Gilles Meghlaoui (gilles.meghlaoui@wanadoo.fr)
Sandrine I Samson (ssamson@spmsd.com)
Emilio Ledesma (emilio.ledesma@hotmail.com)
Margaret Haugh (mhaugh@spmsd.com)

Version: 4 Date: 17 May 2010

Author's response to reviews:

The formatting changes have been made as requested. Please let me know if the figures are acceptable.